Author's response to reviews

Title: Development of a validated Patient Reported Symptom Metric for Pediatric Eosinophilic Esophagitis: Qualitative Methods

Authors:

James P Franciosi (james.franciosi@cchmc.org)
Kevin A Hommel (kevin.hommel@cchmc.org)
Charles W DeBrosse (charles.debrosse@cchmc.org)
Allison B Greenberg (allison.greenberg@cchmc.org)
Alexandria J Greenler (alexandria.greenler@cchmc.org)
J Pablo Abonia (pablo.abonia@cchmc.org)
Marc E Rothenberg (marc.rothenberg@cchmc.org)
Jame W Varni (JVarni@archmail.tamu.edu)

Version: 4 Date: 27 October 2011

Author's response to reviews: see over
Development of a validated Patient Reported Symptom Metric for Pediatric Eosinophilic Esophagitis: Qualitative Methods


Dear members of the Biomed Central editorial staff and reviewers,

We appreciate the thoughtful review of this manuscript and have revised the manuscript in accordance with all of the reviewers’ questions and suggestions. We have carefully reviewed all of your criteria for manuscript submission, criteria for ethical adherence and specifications for our manuscript. The manuscript is approved by all authors. The manuscript has not been published previously (except in the form of an abstract), and is not under consideration (in whole or in part) for publication elsewhere. Please let us know if there is any additional information that we can provide.

Sincerely,

James P. Franciosi, MD, MS, MSCE, FAAP
Cincinnati Children's Hospital Medical Center
Division of Gastroenterology, Hepatology and Nutrition
3333 Burnet Avenue, ML 2010
Cincinnati, OH 45229
James.Franciosi@cchmc.org
Phone: (513) 803-2101
Fax: (513) 736-7805